Author's response to reviews

Title: Comparison of a high and a low intensity smoking cessation intervention in a dentistry setting in Sweden - a randomized trial

Authors:

Eva Nohlert (eva.nohlert@ltv.se)
Ake Tegelberg (ake.tegelberg@ltv.se)
Per Tillgren (per.tillgren@mdh.se)
Pia Johansson (pia.johansson@ki.se)
Andreas Rosenblad (andreas.rosenblad@ltv.se)
Asgeir R Helgason (asgeir.helgason@ki.se)

Version: 10 Date: 23 April 2009

Author's response to reviews:

I have found two things that need to be corrected before publication:

1) In the e-mail addresses pia.johansson@ki.se has another font than the other addresses.

2) In table 4 the CI for the OR for point prevalence is now correctly placed but for continuous abstinence it is still wrong (placed at the LIT row instead of the HIT row).